Author's response to reviews

Title: Anti-inflammatory effect of rosiglitazone is not reflected in expression of NFκB-related genes in peripheral blood mononuclear cells of patients with type 2 diabetes mellitus

Authors:

Marjolijn C.E. Bragt (m.bragt@hb.unimaas.nl)
Jogchum Plat (j.plat@hb.unimaas.nl)
Marco Mensink (marco.mensink@wur.nl)
Patrick Schrauwen (p.schrauwen@hb.unimaas.nl)
Ronald P. Mensink (r.mensink@hb.unimaas.nl)

Version: 2 Date: 24 October 2008

Author's response to reviews:

Dear editor,

Please find enclosed our manuscript entitled “Anti-inflammatory effect of rosiglitazone is not reflected in expression of NFκB-related genes in peripheral blood mononuclear cells of patients with type 2 diabetes mellitus”, which we would like to submit for publication in the BMC Endocrine Disorders.

Our research paper concerns the effect of rosiglitazone, an anti-diabetic medicine, both on plasma inflammatory markers and expression of inflammatory genes in peripheral blood mononuclear cells in type II diabetic patients. It also addresses the influence of high levels of insulin on plasma inflammatory markers. This paper involves inflammation as an important factor in type II diabetes and contributes in understanding the role of PPAR gamma in inflammatory processes. Therefore, we feel that this manuscript adequately fits the scope and readership of BMC Endocrine Disorders.

We declare that we all contributed to the design, execution, and analysis of the study and writing this manuscript. Furthermore, all authors have seen and approved the final version of the manuscript. The data in this manuscript has not been published elsewhere and is not under consideration for publication elsewhere. We also declare that we have no conflicts of interest in connection with this paper.

Please send correspondence to:
Patrick Schrauwen,
Maastricht University,
Department of Human Biology
P.O. Box 616,
6200 MD, Maastricht,
the Netherlands
Telephone: +31-43-3881502
FAX: +31-43-3670976
E-mail: P.Schrauwen@HB.UNIMAAS.NL

Yours sincerely,

Marjolijn C.E. Bragt, MSc